Author’s response to reviews

Title: Lymphoepithelial carcinoma: a case report of a rare tumor of the larynx

Authors:

NAWAL HAMMAS (nawalhammas@gmail.com)
Najib Benmansour (benmansour.dr@gmail.com)
Mohamed Nour-dine EL Alami EL Amine (alami.mn@Gmail.com)
Laila Chbani (chbanil@yahoo.fr)
Hind El Fatemi (elfatemihinde@gmail.com)

Version: 1 Date: 31 May 2017

Author’s response to reviews:

Dear editor,

I have rewritten my manuscript to reduce the level of text overlap